Author's response to reviews

Title: Variable-stiffness colonoscope makes colonoscope easier? A meta-analysis of the efficacy compared variable stiffness colonoscope with standard adult colonoscope

Authors:

Qin Xie (qinqin688418@126.com)
Bin Chen (283024176@qq.com)
Liu Liu (376334117@qq.com)
Tian Hua Gan (23756030@qq.com)

Version: 6 Date: 25 July 2012

Author's response to reviews: see over
Dear Editor,

I resubmit my manuscript to *BMC Gastroenterology* for publication as an “Meta-analysis”. Please find enclosed manuscript with the electronic copy of the full-text in Word format (file name VSC VERSUS SAC.doc)

Firstly, thanks a lot for reviewing. According to the suggestions of the reviewers, I have addressed the comments in the revised manuscript and provided a point-by-point response to the concerns as following:

Reviewer: Javier Sola Vera

Minor essential revisions:
1 I have sought an assistance of fluent English speaking colleague, corrected the misspelled words and grammatical errors carefully.
2 I have accepted the reviewer’s advice and removed the conclusion “probably attenuated the pain of patients at the same time”.

Finally, I want to thank you sincerely for your arduous work and instructive advice and I feel so sorry that so much of your precious time was wasted on my paper revision.

Thanks again for your attention and consideration.

Sincerely yours,

Qin Xie, MD
Department of Gastroenterology
West China Hospital of Sichuan University
No. 37, Guoxue Alley, Chengdu, Sichuan Province, China
TEL:+86-13708047844
E-mail: qinqin688418@126.com